Author's response to reviews

Title: Hamartomas, teratomas and teratocarcinosarcomas of the head and neck: Report of 3 new cases with clinico-pathologic correlation, cytogenetic analysis, and review of the literature

Authors:

Semir Vranic (semir.vranic@gmail.com)
Samuel K Caughron (sam@caughron.com)
Slavisa Djuricic (djurisla@sezampro.yu)
Nurija Bilalovic (nurija@epn.ba)
Sadiq Zaman (sadiqzaman@creighton.edu)
Ismet Suljevic (ismetsul@bih.net.ba)
William M Lydiatt (wmlydiat@unmc.edu)
Jane Emanuel (emanuel@boystown.org)
Zoran Gatalica (zgatalica@pathology.creighton.edu)

Version: 3 Date: 18 October 2008

Author's response to reviews:

October 18, 2008

To:
Dr Mark Todd
Assistant Editor
BMC-series journals


Dear Dr Todd,

We are re-submitting our revised manuscript entitled “Hamartomas, teratomas and teratocarcinosarcomas of the head and neck: Report of three cases and review of the literature” as requested in your email of October 15 2008.

Comments to Reviewer 1:
1. The statement in our abstract has been corrected.
2. The part of discussion has also been modified in line with suggestions of the reviewer.

We hope that these corrections will be satisfactory and the manuscript now can be accepted for publication.
Yours sincerely,

Semir Vranic, MD